Author's response to reviews

Title: Contribution of MRI in the diagnosis of talus skip metastases of calcaneus Ewing’s sarcoma in child: a case report

Authors:

- Hicham Jalal (jhicham2000@hotmail.com)
- Zoubida Belhadj (belhajz@yahoo.fr)
- Hind Enneddam (enneddam-hind@hotmail.com)
- Mohammed Madhar (madmed24@yahoo.fr)
- Tarik Fikry (tfikry@menara.ma)
- Omar Essadki (o.essadki@gmail.com)
- Ahmed Ousehal (a.ousehal@voila.fr)

Version: 4 Date: 19 July 2011

Author's response to reviews:

First of all let me thank you for your considerable efforts and I apologize for exceeding the time limit for corrections.

According to deputy editor Comments to the authors, we've made modifications in the manuscript and all changes have been written in red.

we corrected the manuscript using software (whitesmoke 2011) corrections of grammar, spelling and style mistakes.

And then we made the manuscript checked over by a native English speaker.

We also included the ethnicity of the patient in the abstract and case presentation section of the manuscript so that it conforms fully to the journal style.